Author's response to reviews

Title: Association between physical activity and metabolic syndrome in middle-aged Japanese: a cross-sectional study

Authors:

Junghoon Kim (jh-kim@hotmail.co.jp)
Kai Tanabe (tanabe.kai.gm@u.tsukuba.ac.jp)
Noriko Yokoyama (yokoyama@taiiku.tsukuba.ac.jp)
Hirofumi Zempo (s0837072@u.tsukuba.ac.jp)
Shinya Kuno (kuno@taiiku.tsukuba.ac.jp)

Version: 4 Date: 22 July 2011

Author's response to reviews: see over
We would like to thank you for considering our manuscript entitled “Association between physical activity and metabolic syndrome in middle-aged Japanese: a cross-sectional study” for publication in BMC Public Health.

Thank you for your letter and the comments of July 07, 2011, concerning our manuscript. We believe that we could address and satisfy the criticism of the reviewers. All the answers to the reviewer’s comments are listed on the “Response to Reviews”. In the manuscript, we highlighted change in yellow.

Your consideration of our manuscript is greatly appreciated.

Yours sincerely,

Kindly address and correspondence to:
Shinya Kuno, Ph.D.
Department of Sports Medicine, Graduate School of Comprehensive Human Sciences, University of Tsukuba, Tsukuba, Ibaraki 305-8577, Japan
(TEL) +81-29-853-7335
(FAX) +81 29-853-6407
E-mail: kuno@taiiku.tsukuba.ac.jp